Author's response to reviews

Title: Exercise training for asbestos-related and other dust-related respiratory diseases: a randomised controlled trial

Authors:

Marita T Dale Ms (maritad@uni.sydney.edu.au)
Zoe J McKeough Dr (zoe.mckeough@sydney.edu.au)
Phillip A Munoz Mr (phillip.munoz@sswahs.nsw.gov.au)
Peter Corte Dr (peterc@mail.med.usyd.edu.au)
Peter TP Bye Prof (peter.bye@sydney.edu.au)
Jennifer A Alison Prof (jennifer.alison@sydney.edu.au)

Version: 2 Date: 30 June 2014

Author's response to reviews:

Dear BMC Pulmonary Medicine,

Line numbers have been inserted into the manuscript and the tables have been moved to the end of the document.

Regards, Marita Dale